Author's response to reviews

Title: Diagnosis of rectal cancer by Tissue Resonance Interaction Method

Authors:

Alberto Vannelli (alberto.vannelli@istitutotumori.mi.it)
Luigi Battaglia (luigi.battaglia@istitutotumori.mi.it)
Elia Poiasina (elia.poiasina@istitutotumori.mi.it)
Ermanno Leo (ermanno.leo@istitutotumori.mi.it)

Version: 5 Date: 16 April 2010

Author's response to reviews: see over
Re: MS: 1302989567315735R1
Diagnosis of rectal cancer by Tissue Resonance Interaction Method

Dear Mr. Aulakh,
Thank you for your letter of March 29 saying that our manuscript entitled “Diagnosis of rectal cancer by Tissue Resonance Interaction Method” may be accepted for publication in BMC Gastroenterology.
In the revised manuscript, we have incorporated the ClinicalTrials.gov ID, as requested (shown in blue type; p.4, lines 9-10). With many thanks for your attention.
Sincerely,

Dott. Alberto Vannelli